Author's response to reviews

Title: A process-based framework to guide NP integration into primary healthcare teams.

Authors:

Damien Contandriopoulos (Damien.contandriopoulos@umontreal.ca)
Astrid Brousselle (Astrid.Brousselle@usherbrooke.ca)
Carl-Ardy Dubois (carl.ardy.dubois@umontreal.ca)
Mélanie Perroux (perrouxmelanie@gmail.com)
Marie-Dominique Beaulieu (marie-dominique.beaulieu@umontreal.ca)
Isabelle Brault (isabelle.brault@umontreal.ca)
Kelley Kilpatrick (kelley.kilpatrick@umontreal.ca)
Esther Sansgter-Gormley (egorm@uvic.ca)

Version: 2
Date: 3 July 2014

Author's response to reviews:

Dear Sir or Madam,

As requested, I show clearly the ethics statement in the methodology section. I also numbered lines and pages as requested and detailed the competing interest.

Let me know if any change are needed before sending it to peer-review.

Sincerely,

Mélanie Perrou